Author’s response to reviews

Title: Lipid Metabolism Enzyme ACSVL3 Supports Glioblastoma Stem Cell Maintenance and Tumorigenicity

Authors:

Peng Sun (sun@kennedykrieger.org)
Shuli Xia (xia@kennedykrieger.org)
Kil Sung Yang (kyang@jhsph.edu)
Paul Watkins (watkins@kennedykrieger.org)
John Laterra (laterra@kennedykrieger.org)

Version: 2 Date: 4 October 2013

Author’s response to reviews:

Dear Editor,

We revised the manuscript per the instructions send from the BMC cancer editorial office.

We reformatted the abstract to include the "Background, Methods, Results and Conclusions” sections.

We moved the authors' contribution to the correct place.

We added the "Competing interests" and "Acknowledgements" section at the end of the manuscript.

In the Methods section (end of first paragraph), we clarified the use of human samples for this study and indicated that it is designated by Johns Hopkins IRB as "non-human subjects research”.

We hope these changes meet all the requirement of the submission. Thanks.

Sincerely,

John Laterra